19448007, 2023, 4, Dov



Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2022GL102077

Key Points:

- Future Arctic climate change in winter strongly depends on the strength of the increase in poleward ocean heat transport (OHT)
- Arctic climate change in the NEMO-family climate models is much more pronounced
- NEMO-family climate models have larger increases in poleward OHT

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

Q. Shu and F. Qiao, shuqi@fio.org.cn; qiaofl@fio.org.cn

Citation:

Pan, R., Shu, Q., Wang, Q., Wang, S., Song, Z., He, Y., & Qiao, F. (2023). Future Arctic climate change in CMIP6 strikingly intensified by NEMO-family climate models. *Geophysical Research Letters*, 50, e2022GL102077. https://doi. org/10.1029/2022GL102077

Received 11 NOV 2022 Accepted 20 JAN 2023

Author Contributions:

Conceptualization: Qi Shu, Qiang Wang, Fangli Qiao Investigation: Rongrong Pan, Shizhu Wang, Zhenya Song, Yan He Methodology: Qi Shu Visualization: Rongrong Pan Writing – original draft: Rongrong Pan, Qi Shu, Qiang Wang Writing – review & editing: Rongrong Pan, Qi Shu, Qiang Wang, Fangli Qiao

© 2023 The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

Future Arctic Climate Change in CMIP6 Strikingly Intensified by NEMO-Family Climate Models

Rongrong Pan^{1,2,3}, Qi Shu^{1,2,3}, Qiang Wang⁴, Shizhu Wang^{1,2,3}, Zhenya Song^{1,2,3}, Yan He^{1,2,3}, and Fangli Qiao^{1,2,3}

¹First Institute of Oceanography, Key Laboratory of Marine Science and Numerical Modeling, Ministry of Natural Resources, Qingdao, China, ²Laboratory for Regional Oceanography and Numerical Modeling, Pilot National Laboratory for Marine Science and Technology, Qingdao, China, ³Shandong Key Laboratory of Marine Science and Numerical Modeling, Qingdao, China, ⁴Alfred Wegener Institute Helmholtz Centre for Polar and Marine Research (AWI), Bremerhaven, Germany

Abstract Climate change in the Arctic has substantial impacts on human life and ecosystems both within and beyond the Arctic. Our analysis of CMIP6 simulations shows that some climate models project much larger Arctic climate change than other models, including changes in sea ice, ocean mixed layer, air-sea heat flux, and surface air temperature in wintertime. In particular, dramatic enhancement of Arctic Ocean convection down to a few hundred meters is projected in these models but not in others. Interestingly, these models employ the same ocean model family (NEMO) while the choice of models for the atmosphere and sea ice varies. The magnitude of Arctic climate change is proportional to the strength of the increase in poleward ocean heat transport, which is considerably higher in this group of models. Establishing the plausibility of this group of models with high Arctic climate sensitivity to anthropogenic forcing is imperative given the implied ramifications.

Plain Language Summary Arctic climate is projected to change dramatically in the future based on the latest climate models. However, there is a large discrepancy between these climate models. We find that simulated Arctic climate change tends to be much larger in models using the community ocean model "NEMO" as their ocean component. Future Arctic climate change in winter strongly depends on the strength of the increase in poleward ocean heat transport, which is considerably higher in the NEMO-family climate models. There are some indications that this group of models can better reproduce the observed winter sea ice decline and mixed layer depth in the Barents Sea in their historical simulations. We suggest that investigating the model physics in the NEMO-family climate models in comparison with other models may ultimately help to reduce the overall uncertainty in climate change projections.

1. Introduction

The Arctic is very susceptible to global climate change. Its surface air temperature (SAT) is warming at a rate 2–4 times higher than the global average starting from the 1970s (Bekryaev et al., 2010; Holland & Bitz, 2003; Rantanen et al., 2022; Serreze & Barry, 2011). The phenomenon of Arctic amplification has been found not only in historical observations and climate model projection simulations, but also in proxy reconstructions of paleo climate changes (Pithan & Mauritsen, 2014; Renssen et al., 2009). Warming amplification has also been detected in the ocean component of the Arctic based on analysis of climate models (Shu et al., 2022). The sea ice decay associated with Arctic warming is a crucial indicator of both Arctic and global climate changes (Notz & Stroeve, 2016; Polyakov et al., 2010; Shu et al., 2022). Arctic sea ice extent (SIE) in September has halved since the late 1970s (Perovich et al., 2017) and the location of the winter sea ice edge has also considerably retreated, potentially affecting weather and climate across the Northern Hemisphere (Cohen et al., 2014; Coumou et al., 2018; Overland et al., 2016; Screen et al., 2018).

Atmosphere-ocean-sea ice interactions play an important role in Arctic climate change. The presence of Arctic sea ice limits the vertical mixing in the upper ocean and maintains a cold halocline (Fer, 2009; Polyakov et al., 2017), which is an insulating barrier between the underlying warm Atlantic Water layer and the mixed layer and sea ice above. However, with the rapid increase in Arctic temperatures and the melting of sea ice, the Arctic Ocean stratification is changing. The reduction in sea ice cover has promoted stronger vertical mixing in the upper ocean (Rainville & Woodgate, 2009; Tsamados et al., 2014) and weakened the halocline stratification in the Arctic Eurasian Basin (Polyakov, Rippeth, et al., 2020; Wang & Danilov, 2022). These ocean changes could contribute

to the ventilation of the Atlantic Water layer and enhance sea ice basal melting (Polyakov et al., 2017; Polyakov, Alkire, et al., 2020). A positive trend in the poleward ocean heat transport (OHT) to the Arctic Ocean by Atlantic inflows over the past decades has been found, and it has greatly influenced the Arctic climate (Årthun et al., 2012; Barton et al., 2018; Tsubouchi et al., 2021; Wang et al., 2020). Arctic Atlantification was observed in the Barents Sea and the eastern Eurasian Basin (Ingvaldsen et al., 2021; Polyakov et al., 2017). Climate model simulations indicate that persistent increases in Atlantic Water heat inflow to the Arctic Ocean in a warming climate will sustain the Arctic Ocean amplification (Shu et al., 2022) and feed the Arctic atmospheric amplification as well (Beer et al., 2020; Shu et al., 2022; van der Linden et al., 2019). However, in a future warming climate, the Barents Sea will lose cooling efficiency due to stronger atmospheric warming in this region, and the remaining ocean heat will become an important source of ocean warming in the Arctic deep basin, with strong impacts on ocean stratification, winter sea ice cover and atmosphere warming in the Arctic interior (Shu et al., 2021, 2022).

Climate models are key tools for understanding future climate changes (Jahn, 2018; Landrum & Holland, 2020; Screen et al., 2018), although they are often subject to large uncertainties in the Arctic, such as for the simulated Arctic warming (Latonin et al., 2021; Pithan & Mauritsen, 2014; Smith et al., 2019), sea ice decline (Docquier et al., 2019; Li et al., 2017; Notz & Community, 2020), and Arctic Ocean hydrography and stratification (Heuzé et al., 2023; Khosravi et al., 2022; Muilwijk et al., 2022; Shu et al., 2019). Future changes in the Arctic climate will have significant impacts on human life and marine ecosystems (Chambault et al., 2022; Ingvaldsen et al., 2021). Therefore, it is required to improve the understanding of the extent and influencing factors of the Arctic climate sensitivity in state-of-the-art climate models.

Here, based on the simulations of the Coupled Model Intercomparison Project Phase 6 (CMIP6), we investigate the processes of Arctic atmosphere-ocean-sea ice interactions in a warming climate. We find that the winter ocean surface mixed layer depth (MLD), sea ice cover, sea surface heat flux (SSHF), and SAT in the Arctic are projected to change dramatically in the future, but their changes have great discrepancies across CMIP6 models, especially in the Eurasian Basin. The discrepancies are significantly correlated between these climate variables, and can be largely explained by the difference in the simulated poleward OHT from the North Atlantic, which interestingly has a strong dependence on the ocean model component used in the climate models. Specifically, we find that the large group of climate models with NEMO (Nucleus for European Modelling of the Ocean) as their ocean model component tend to simulate higher poleward OHT and thus much more rapid Arctic climate change compared to other climate models.

2. Materials and Methods

The first realizations of 23 CMIP6 coupled models listed in Table S1 in Supporting Information S1 are used. These are the models that provided all the required variables. The simulations for the period from 1950 to 2014 are from their historical experiments. The future projections are driven by external forcing based on the Shared Socioeconomic Pathway 585 (SSP5-8.5) scenario, which represents the high end of the range of future pathways with an effective radiative forcing of 8.5 W m⁻² in 2100 (O'Neill et al., 2016; Riahi et al., 2017).

To assess the simulations of sea ice in CMIP6 climate models in the historical period, we use satellite observations of SIE from the National Snow and Ice Data Center (NSIDC; Meier et al., 2021). We compare the simulated OHT in CMIP6 historical simulations with the estimate from the Ocean ReAnalysis System 5 (ORAS5; Zuo et al., 2019). The fifth generation ECMWF reanalysis (EAR5; Hersbach et al., 2020) and Polar science center Hydrographic Climatology (PHC; Steele et al., 2001) are also used in this study to evaluate the simulations of Arctic surface air warming and the climatology of ocean surface MLD, respectively.

In this study, the ocean surface MLD is defined as the depth where the potential density is larger than the surface potential density by 0.1 kg/m^3 (Peralta-Ferriz & Woodgate, 2015). The SIE is calculated as the sum of grid cell areas where sea ice concentration (SIC) is larger than 15%. The OHT through each Arctic Ocean gateway is calculated using monthly seawater potential temperature and ocean velocity following Shu et al. (2022) as:

$$OHT(v,T) = \rho_0 c_p \int_{-H(\lambda)}^0 \int_{\lambda_1(z)}^{\lambda_2(z)} v(T - T_{ref}) d\lambda dz$$
(1)

where ρ_0 is the density of sea water, c_p is the specific heat capacity of sea water, v is ocean velocity locally perpendicular to the section of each gateway, T is potential temperature in the section, T_{ref} is reference temperature set



Geophysical Research Letters

10.1029/2022GL102077



Figure 1. Arctic future climate changes. The changes of winter (January to March) average (a) ocean mixed layer depth (MLD), (c) sea surface heat flux (SSHF), (e) sea ice concentration (SIC), and (g) surface air temperature (SAT) for the period of 2081–2100 relative to the period of 1979–2020 based on CMIP6 multimodel mean results. Their intermodel spreads (one standard deviation) are shown in (b), (d), (f), and (h), respectively. The black box in (a) represents the Eurasian Basin used in Figures 2 and 4. Positive SSHF changes in (c) indicate enhanced ocean surface heat loss.

to 0°C in this study, H is water depth, and λ is the distance along the gateway transect. OHT is calculated along the zigzag grid lines on the native model grids to largely guarantee heat conservation and avoid ambiguities associated with a regridded land/sea mask.

3. Results

3.1. Arctic Future Climate Change in CMIP6

Under the SSP5-8.5 scenario, dramatic future changes in Arctic climate are projected by CMIP6 models (Figure 1). The depth of the ocean surface mixed layer of the Arctic Ocean has important impacts on the heat transfer between the ocean, sea ice, and atmosphere (Sirevaag et al., 2011; Steele et al., 2011). CMIP6 multimodel mean (MMM) results show that winter MLD in the eastern Eurasian Basin will deepen markedly in the future. Relative to the period of 1979–2020, the winter MLD averaged over the whole Eurasian Basin is projected to be deeper by >50 m in the period of 2081–2100 (the long-term future period considered in this study), while regionally the increase in the MMM MLD along the Eurasian continental slope can reach >100 m (Figure 1a). The deepening of the MLD is related to stronger winter convection represented by more sea surface heat release (Figure 1c). The largest increase in winter SSHF is located along the inflow pathways of Atlantic and Pacific waters. In the Atlantic sector, the strong increase in the MLD of >100 m is colocated with the dramatic increase in the SSHF of >80 W m⁻² in the northern Kara Sea and along the continental slope of the Eurasian Basin (Figures 1a and 1c). The regional decrease in MLD and SSHF in the southwestern Barents Sea reflects the reduction in the ocean cooling efficiency in this region in a warming climate (Shu et al., 2021; Skagseth et al., 2020).

The increase in ocean surface heat release is associated with the reduction in the sea ice cover in a warming climate, as indicated by the spatial correlation between the changes in SSHF and SIC (compare Figures 1c



Figure 2. The relationship between the future changes of main climate variables in CMIP6 models. Scatter plots of future changes: (a) winter mixed layer depth (MLD) versus winter sea surface heat flux (SSHF), (b) winter SSHF versus winter sea ice extent (SIE), (c) winter SIE versus annual mean ocean heat transport (OHT) through the Barents Sea Opening (BSO), (d) winter MLD versus annual mean OHT through the BSO, (e) ocean temperature along the BSO versus global mean surface air temperature (SAT), and (f) OHT through the BSO versus global mean SAT. The MLD, SSHF, and SIE are calculated over the Eurasian Basin indicated by the black box in Figure 1. The black lines show the linear fit. The changes are the differences between the periods of 2081–2100 and 1979–2020. CMIP6 models with NEMO as their ocean component are marked by triangles, and other models are marked by circles.

and 1e). For the period of 2081–2100, the winter SIC decreases by >30% in the Arctic Ocean, with the greatest reduction of >60% in the Eurasian Basin, Chukchi Sea, northern Barents Sea, and Kara Sea (Figure 1e). Winter sea ice decline weakens the insulating effect of sea ice in air-sea heat exchange. The winter SAT in the Arctic will rise dramatically (Figure 1g), which partially can be attributed to the increase in ocean heat release to the atmosphere. The winter SAT averaged over the Arctic will rise by >14°C at the end of the 21st century. The SAT increase in the midterm future (2041–2060) clearly demonstrates the direct effect of the ocean surface heat release along the Atlantic Water inflow in warming the atmosphere, as the SAT increase is intensified over the Kara Sea and Eurasian Basin (Figure S1 in Supporting Information S1). The spatial pattern of the SAT change becomes more uniform in the Arctic at the end of the 21st century (Figure 1g), which is linked to the fact that the winter sea ice decline and SSHF increase in the central Arctic are more apparent in the long-term future (Figures 1c, 1e, and Figures S1c, S1e in Supporting Information S1).

The spatial patterns of the future wintertime climate change in the Arctic indicate that the ocean, sea ice, and atmosphere in the Arctic strongly interact with each other, and the ocean heat convergence to the Arctic Ocean from lower latitudes is a key player in driving Arctic climate change. The latter implies that the representation of the ocean processes in the CMIP6 models can strongly influence the simulated Arctic climate change, which will be addressed below by analyzing individual models.

3.2. Different Arctic Climate Sensitivity

The changes in MLD, SSHF, SIC, and SAT in CMIP6 models have a large intermodel spread (one standard deviation) (Figure 1 and Figure S1 in Supporting Information S1), which indicates that CMIP6 models have a great difference in Arctic climate sensitivity. For example, winter MLD in 10 models (CanESM5, CanESM5-CanOE, CMCC-CM2-SR5, CNRM-CM6-1, CNRM-ESM2-1, EC-Earth3, EC-Earth3-Veg, HadGEM3-GC31-LL, IPSL-CM6A-LR, and UKESM1-0-LL) will deepen markedly in the Eurasian Basin (Figure S2 in Supporting Information S1). In these models, the winter MLD increases by >200–300 m in the eastern Eurasian Basin, while MLD increase is nearly absent in this region in other models.

We find that the climate change signals of winter MLD, SSHF, and SIE in the Eurasian Basin are significantly correlated among the models (Figure 2). The correlation coefficients of the SSHF changes with the MLD and SIE changes are 0.78 and -0.93 (Figures 2a and 2b), respectively, and both exceed the 99% confidence level. In winter, lower SIE with larger open water areas allows stronger surface cooling as indicated by larger sea surface heat release, which causes stronger winter convection and deeper MLD. Meanwhile, deeper MLD tends to bring more oceanic heat upwards to enhance sea ice basal melting. Therefore, the physical process of atmosphere-ocean-sea ice interaction can explain the significant correlations between the model spreads of the simulated MLD, SSHF, and SIE.

As mentioned above, the largest changes in MLD, SSHF, and SIC in the Atlantic sector follow the pathways of Atlantic Water inflow (Figure 1 and Figure S1 in Supporting Information S1). This indicates that future Arctic climate changes are closely related to the increased OHT into the Arctic Ocean, which can cause ocean warming, sea ice decline, and increase in SSHF and MLD in previously ice-covered areas (Årthun et al., 2017; Docquier & Koenigk, 2021; Shu et al., 2021, 2022). Therefore, significant correlations can be found between the future changes of the winter SIE and MLD in the Eurasian Basin and the future changes of the annual mean OHT through the Barents Sea Opening (BSO; Figures 2c and 2d). The future increase in net OHT through the BSO is much more prominent than that through the Fram Strait (Figure S3 in Supporting Information S1). The future changes in OHT through the BSO in the CMIP6 models have a great discrepancy with a range from about 0 to 170 TW (Figures 2c and 2d), so the future changes in Arctic climate are very different between the models.

3.3. Arctic Climate Sensitivity Linked to Ocean Component Models

The magnitudes of simulated climate changes can be influenced by many factors, such as natural internal variability, model numerics, parameterizations, model resolutions. Here, we find that the future Arctic climate projections are closely linked to the ocean models used in different climate models. Among the 23 climate models analyzed in this study, 11 models employ the NEMO ocean model or modified versions based on it (Table S1 in Supporting Information S1). It is obvious that the NEMO-family climate models tend to simulate larger climate changes with more pronounced increase in MLD and SSHF and larger reduction in SIE in the Eurasian Basin in accordance with stronger increase in BSO heat inflow (Figures 2a–2d, where NEMO-family climate models are marked with triangles). The aforementioned 10 models with regional MLD increase of >200 m in the eastern Eurasian Basin all belong to the NEMO family (Figure S2 and Table S1 in Supporting Information S1). Other model components of the climate models and detailed model configurations can also influence the simulated Arctic climate changes as shown by the different results within the climate models of the NEMO family, but our analysis suggests that it is highly possible that the choice of ocean models have the largest impact on the Arctic climate change projections.

The striking impact of the ocean models on the simulated Arctic climate changes is illustrated by comparing the changes in winter MLD, SSHF, SIC, SIE, and SAT between the NEMO-family climate models and the non-NEMO models in Figures 3 and 4. There is a strong contrast in the winter MLD changes between the two model groups (Figures 3a and 3b). The NEMO-family climate models simulate a pronounced mixed layer deepening of >200 m

10.1029/2022GL102077



Geophysical Research Letters



Figure 3. Arctic future climate changes in two groups of CMIP6 models. The changes of winter (January to March) (a, b) ocean mixed layer depth (MLD), (c, d) sea surface heat flux (SSHF), (e, f) sea ice concentration (SIC), and (g, h) surface air temperature (SAT) for the period of 2081–2100 relative to the period of 1979–2020. (a), (c), (e), and (g) are multimodel mean (MMM) results of the climate models employing NEMO. (b), (d), (f), and (h) are MMM results of other climate models.

in the eastern Eurasian Basin (Figure 3a), while the non-NEMO models only obtain MLD increases of a few tens of meters in the northern Barents and Kara Seas (Figure 3b). The MMM MLD increase in the Eurasian Basin in CMIP6 projections can be mainly attributed to the NEMO-family climate models (Figure 4a). The increase in the SSHF in the eastern Eurasian Basin can reach >150 W m⁻² averaged over the NEMO-family climate models, about twice the magnitude of the non-NEMO models (Figures 3c and 3d).

The projections of Arctic sea ice also differ dramatically between these two groups of models for SIC (Figures 3e and 3f) and SIE (Figure 4b). In the NEMO-family climate models, the strong reduction of SIC in the southern Eurasian Basin and Chukchi Sea implies that these areas will be nearly ice-free in winter at the end of the 21st century, while in the non-NEMO models the SIC reduction is only about 50% in these areas (Figures 3e and 3f). In these two groups of models, the winter SIE in the Eurasian Basin is 2.5×10^5 km² (80% reduction) and

 6.5×10^5 km² (50% reduction) at the end of this century, respectively (Figure 4b).

A large discrepancy in SAT projections between these two model groups is also clear. In 2081–2100, the winter SAT will increase by >24°C in the central Arctic in the NEMO-family climate models, but <18°C in the non-NEMO models (Figures 3g and 3h). Averaged across the Arctic, the increase in the winter SAT at the end of the 21st century is about 5°C higher in the NEMO-family climate models (Figure 4c).

The intriguing difference in the simulated Arctic climate changes between the two model groups is associated with their great discrepancy in the OHT through the BSO (Figure 4d). On average, both model groups simulate positive trends in the OHT through the BSO in a warming climate, but the OHT in the NEMO-family climate models is nearly twice as much as in the non-NEMO models at the end of the 21st century (Figure 4d). For the Fram Strait, the simulation discrepancy in the future change of the net OHT between the two model groups is relatively small compared with the BSO (Figure S3 in Supporting Information S1). The discrepancy in the OHT through the Bering Strait between the two model groups (Figure S3 in Supporting Information S1) is consistent





Figure 4. Time series of Arctic climate variables and ocean heat transport (OHT). (a) Winter (January to March) ocean mixed layer depth (MLD) and (b) winter sea ice extent (SIE) in the Eurasian Basin (EB). (c) Winter surface air temperature (SAT) anomaly in the Arctic relative to the period of 1951–1970. (d) Annual mean net OHT through the Barents Sea Opening (BSO). (e) Winter SIE in the Barents Sea (BS) in the historical simulations and observations. (f) Annual mean OHT through the BSO during 1979–2018. The green line in (f) is the estimated OHT through the BSO in the period of 1997–2008 from Smedsrud et al. (2010). The magenta dashed line in (f) is the estimated OHT through the BSO in the period of 1993–2010 based on the average of 10 different reanalysis data sets from Uotila et al. (2019).

with their difference in winter sea ice decline in the Pacific sector of the Arctic Ocean (Figures 3e and 3f). The larger increases in OHT through the BSO and Bering Strait cause more rapid Arctic Ocean warming in the NEMO-family climate models. The Arctic Ocean Amplification factor, the ratio of the warming rate of the Arctic Ocean to that of the global ocean (considering the whole ocean depth; Shu et al., 2022) is 3.7 and 2.1 in the NEMO and non-NEMO family of climate models, respectively.

The positive trend in OHT through the BSO is mainly caused by the increases in ocean temperature at the BSO in a warming climate (Shu et al., 2022), which is the case for both the model groups (Figures S4 and S5 in Supporting Information S1). Significant correlations are found between the future changes of ocean temperature and OHT at the BSO and the future change of the global mean SAT (Figures 2e and 2f). Despite the fact that all models are driven by the same anthropogenic forcing, the NEMO-family climate models tend to simulate much stronger global warming, which is associated with larger poleward OHT. Two non-NEMO climate models CESM2 and CESM2-WACCM also have high global climate sensitivity to anthropogenic forcing (Zelinka et al., 2020) and thus relatively large increase in global mean SAT in a warming climate, but they have weak increases in poleward OHT (Figure 2f), thus weak sea ice decline and small changes in MLD (Figures 2c and 2d).

4. Discussion

CMIP6 MMM results indicate that the Arctic will experience rapid climate changes in the future under the SSP5-8.5 scenario, especially in the Eurasian Basin, where winter atmosphere-ocean-sea ice interaction will become stronger in a warming climate. However, CMIP6 models show great discrepancy in the magnitude of the projected Arctic climate change. Our study reveals that the magnitudes of the Arctic climate changes in winter, including sea ice cover, ocean surface mixed layer, sea surface heat loss, and air temperature, are significantly correlated with future changes in poleward ocean heat convergence. We also find that the climate models with NEMO as their ocean component tend to project larger increase in OHT into the Arctic Ocean, which further leads to greater Arctic climate changes in winter. So, a large part of the intermodel spread in Arctic climate sensitivity stems from the choice of ocean component in coupled climate models. The reasons for this ocean model dependence need further investigation, and warrant a concerted effort in the model development community. For example, the role of vertical mixing schemes employed in coupled climate models should be better understood. NEMO employs the TKE turbulent closure scheme (Madec et al., 1998), while the K-Profile Parameterization vertical mixing scheme (Large et al., 1994) is often used in other ocean models.

Over the past several decades, increased OHT through the BSO induced rapid sea ice decline in the Barents Sea in winter, which is one of the most prominent signals of climate change in the Arctic. The comparison between observations and the two groups of CMIP6 climate models for the winter SIE in the Barents Sea (Figure 4e) indicates that the non-NEMO models slightly overestimate the SIE mean state and underestimate the SIE trend, while the NEMO-family climate models are closer to observations. We also use the OHT through the BSO in the ORAS5 reanalysis and the estimate from a set of reanalysis data sets (Uotila et al., 2019) to evaluate CMIP6 climate models. The results suggest that non-NEMO models might slightly underestimate the mean OHT and its upward trend in the historical simulations, while NEMO-family climate models might have slight overestimations (Figure 4f). We note that these reanalysis data sets are also based on simulations and might be subject to uncertainties. The difference of Arctic surface warming trends during 1979-2021 between the two model groups is consistent with the difference in OHT, with the NEMO-family climate models overestimating the air temperature trend in the Barents Sea and non-NEMO climate models underestimating the trend (Figure S6 in Supporting Information S1). MLDs from these two model groups are also compared with that derived from PHC3.0 (Figure S7 in Supporting Information S1), indicating that the NEMO-family climate models are slightly better in simulating winter MLD. The above comparisons of sea ice and MLD simulations indicate that the NEMO-family climate models may better project future Arctic climate change, while this is somehow inconclusive considering the model performance in the representation of OHT and air warming in the historical period.

Studies using previous-generation climate models indicate that deep convection is likely to emerge in the Arctic Ocean in a warming climate (Lique & Thomas, 2018; Lique et al., 2018). Here, we show that in CMIP6 models deep convection of a few hundred meters may emerge within this century under the scenario of SSP5-8.5 only in NEMO-family climate models. Considering the surprisingly large differences in simulated Arctic climate change between the NEMO-family climate models and non-NEMO climate models, even when using multiple models from CMIP6 to assess future Arctic climate, the MMM results can be strongly influenced by the availability of required model data from different groups of models (i.e., how many models from each group are used in calculating the MMM). In an extreme case, if a single climate model is used to assess the Arctic climate change in the future, the findings, including the future change in Arctic Ocean surface mixed layer and sea ice cover, will be radically different depending on which model family is used. Effort to reduce the large intrinsic model uncertainties in simulating Arctic climate is therefore needed. Our findings about the strong dependence of Arctic climate changes on ocean models imply a possibility to identify and reduce a considerable part of climate model intrinsic uncertainties.

Data Availability Statement

The CMIP6 models used in this study are listed in Table S1 in Supporting Information S1. The CMIP6 data sets can be downloaded from https://esgf-node.llnl.gov/search/cmip6/. Satellite-observed SIC data set are from NSIDC at https://doi.org/10.7265/efmz-2t65. The ocean reanalysis data (ORAS5) are from https://doi.org/10.24381/cds.67e8eeb7. The ERA5 is from https://doi.org/10.24381/cds.f17050d7. The PHC3.0 is from http://psc.apl.washington.edu/nonwp_projects/PHC/Data3.html.



Acknowledgments

Q.S. was supported by the Chinese Natural Science Foundation (41941012, 41821004, and 42276253), Shandong Provincial Natural Science Foundation (ZR2022JQ17), and the Taishan Scholars Program. Q.W. was supported by the German Helmholtz Climate Initiative REKLIM (Regional Climate Change and Human) and by the German Federal Ministry for Education and Research with funding number 03F0889A.

References

- Årthun, M., Eldevik, T., Smedsrud, L., Skagseth, Ø., & Ingvaldsen, R. (2012). Quantifying the influence of Atlantic heat on Barents Sea ice variability and retreat. *Journal of Climate*, 25(13), 4736–4743. https://doi.org/10.1175/JCLI-D-11-00466.1
- Årthun, M., Eldevik, T., Viste, E., Drange, H., Furevik, T., Johnson, H. L., & Keenlyside, N. S. (2017). Skillful prediction of northern climate provided by the ocean. *Nature Communications*, 8(1), 15875. https://doi.org/10.1038/ncomms15875
- Barton, B. I., Lenn, Y.-D., & Lique, C. (2018). Observed Atlantification of the Barents Sea causes the polar front to limit the expansion of winter sea ice. *Journal of Physical Oceanography*, 48(8), 1849–1866. https://doi.org/10.1175/JPO-D-18-0003.1
- Beer, E., Eisenman, I., & Wagner, T. J. W. (2020). Polar amplification due to enhanced heat flux across the halocline. *Geophysical Research Letters*, 47, e2019GL086706. https://doi.org/10.1029/2019GL086706
- Bekryaev, R., Polyakov, I. V., & Alexeev, V. (2010). Role of polar amplification in long-term surface air temperature variations and modern Arctic warming. *Journal of Climate*, 23(14), 3888–3906. https://doi.org/10.1175/2010JCLI3297.1
- Chambault, P., Kovacs, K. M., Lydersen, C., Shpak, O., Teilmann, J., Albertsen, C. M., & Heide-Jørgensen, M. P. (2022). Future seasonal changes in habitat for Arctic whales during predicted ocean warming. *Science Advances*, 8(29), eabn2422. https://doi.org/10.1126/sciady.abn2422
- Cohen, J., Screen, J. A., Furtado, J. C., Barlow, M., Whittleston, D., Coumou, D., et al. (2014). Recent Arctic amplification and extreme midlatitude weather. *Nature Geoscience*, 7(9), 627–637. https://doi.org/10.1038/ngeo2234
- Coumou, D., Di Capua, G., Vavrus, S., Wang, L., & Wang, S. (2018). The influence of Arctic amplification on mid-latitude summer circulation. *Nature Communications*, 9(1), 2959. https://doi.org/10.1038/s41467-018-05256-8
- Docquier, D., Grist, J., Roberts, M., Roberts, C., Semmler, T., Ponsoni, L., et al. (2019). Impact of model resolution on Arctic sea ice and North Atlantic Ocean heat transport. *Climate Dynamics*, 53(7–8), 4989–5017. https://doi.org/10.1007/s00382-019-04840-y
- Docquier, D., & Koenigk, T. (2021). A review of interactions between ocean heat transport and Arctic sea ice. *Environmental Research Letters*, 16(12), 123002. https://doi.org/10.1088/1748-9326/ac30be
- Fer, I. (2009). Weak vertical diffusion allows maintenance of cold halocline in the central Arctic. Atmospheric and Oceanic Science Letters, 2(3), 148–152. https://doi.org/10.1080/16742834.2009.11446789
- Hersbach, H., Bell, B., Berrisford, P., Hirahara, S., Horányi, A., Muñoz-Sabater, J., et al. (2020). The ERA5 global reanalysis. *Quarterly Journal of the Royal Meteorological Society*, 146(730), 1999–2049. https://doi.org/10.1002/qj.3803
- Heuzé, C., Zanowski, H., Karam, S., & Muilwijk, M. (2023). The deep Arctic Ocean and Fram Strait in CMIP6 models. Journal of Climate, 1–68. https://doi.org/10.1175/JCLI-D-22-0194.1
- Holland, M. M., & Bitz, C. M. (2003). Polar amplification of climate change in coupled models. *Climate Dynamics*, 21(3–4), 221–232. https://doi.org/10.1007/s00382-003-0332-6
- Ingvaldsen, R. B., Assmann, K. M., Primicerio, R., Fossheim, M., Polyakov, I. V., & Dolgov, A. V. (2021). Physical manifestations and ecological implications of Arctic Atlantification. *Nature Reviews Earth & Environment*, 2(12), 874–889. https://doi.org/10.1038/s43017-021-00228-x
- Jahn, A. (2018). Reduced probability of ice-free summers for 1.5°C compared to 2°C warming. *Nature Climate Change*, 8(5), 409–413. http s://doi.org/10.1038/s41558-018-0127-8
- Khosravi, N., Wang, Q., Koldunov, N., Hinrichs, C., Semmler, T., Danilov, S., & Jung, T. (2022). The Arctic Ocean in CMIP6 models: Biases and projected changes in temperature and salinity. *Earth's Future*, 10, e2021EF002282. https://doi.org/10.1029/2021EF002282
- Landrum, L., & Holland, M. M. (2020). Extremes become routine in an emerging new Arctic. *Nature Climate Change*, 10(12), 1108–1115. https://doi.org/10.1038/s41558-020-0892-z
- Large, W. G., McWilliams, J. C., & Doney, S. C. (1994). Oceanic vertical mixing: A review and a model with a nonlocal boundary layer parameterization. *Reviews of Geophysics*, 32(4), 363–403. https://doi.org/10.1029/94RG01872
- Latonin, M. M., Bashmachnikov, I. L., Bobylev, L. P., & Davy, R. (2021). Multi-model ensemble mean of global climate models fails to reproduce early twentieth century Arctic warming. *Polar Science*, 30, 100677. https://doi.org/10.1016/j.polar.2021.100677
- Li, D., Zhang, R., & Knutson, T. R. (2017). On the discrepancy between observed and CMIP5 multi-model simulated Barents Sea winter sea ice decline. *Nature Communications*, 8(1), 14991. https://doi.org/10.1038/ncomms14991
- Lique, C., Johnson, H. L., & Plancherel, Y. (2018). Emergence of deep convection in the Arctic Ocean under a warming climate. *Climate Dynamics*, 50(9), 3833–3847. https://doi.org/10.1007/s00382-017-3849-9
- Lique, C., & Thomas, M. (2018). Latitudinal shift of the Atlantic Meridional Overturning Circulation source regions under a warming climate. *Nature Climate Change*, 8(11), 1013–1020. https://doi.org/10.1038/s41558-018-0316-5
- Madec, G., Delécluse, P., Imbard, M., & Lévy, C. (1998). OPA 8.1 ocean general circulation model reference manual. Notes Du Pôle de Mode 'lisation, Laboratoire d'oce 'anographie Dynamique et de Climatologie. Institut Pierre Simon Laplace Des Sciences de l'environnement Global, (Vol. 11).
- Meier, W. N., Fetterer, F., Windnagel, A. K., & Stewart, J. S. (2021). NOAA/NSIDC climate data record of passive microwave sea ice concentration, Version 4 [Dataset]. National Snow and Ice Data Center. https://doi.org/10.7265/efmz-2t65
- Muilwijk, M., Nummelin, A., Heuzé, C., Polyakov, I. V., Zanowski, H., & Smedsrud, L. H. (2022). Divergence in climate model projections of future Arctic Atlantification. Journal of Climate, 1–53. https://doi.org/10.1175/JCLI-D-22-0349.1
- Notz, D., & Community, S. (2020). Arctic sea ice in CMIP6. Geophysical Research Letters, 47, e2019GL086749. https://doi.org/10.1029/ 2019GL086749
- Notz, D., & Stroeve, J. (2016). Observed Arctic sea-ice loss directly follows anthropogenic C₀2 emission. *Science*, *35*4(6313), 747–750. http s://doi.org/10.1126/science.aag2345
- O'Neill, B., Tebaldi, C., Vuuren, D., Eyring, V., Friedlingstein, P., Hurtt, G., et al. (2016). The Scenario Model Intercomparison Project (ScenarioMIP) for CMIP6. Geoscientific Model Development Discussions, 9, 3461–3482. https://doi.org/10.5194/gmd-2016-84
- Overland, J. E., Dethloff, K., Francis, J. A., Hall, R. J., Hanna, E., Kim, S.-J., et al. (2016). Nonlinear response of mid-latitude weather to the changing Arctic. *Nature Climate Change*, 6(11), 992–999. https://doi.org/10.1038/nclimate3121
- Peralta-Ferriz, C., & Woodgate, R. A. (2015). Seasonal and interannual variability of pan-Arctic surface mixed layer properties from 1979 to 2012 from hydrographic data, and the dominance of stratification for multiyear mixed layer depth shoaling. *Progress in Oceanography*, 134, 19–53. https://doi.org/10.1016/j.pocean.2014.12.005
- Perovich, D. K., Gerland, S., Hendricks, S., Meier, W. N., Nicolaus, M., & Tschudi, M. (2017). The arctic: Sea ice cover. American Meteorological Society.
- Pithan, F., & Mauritsen, T. (2014). Arctic amplification dominated by temperature feedbacks in contemporary climate models. *Nature Geoscience*, 7(3), 181–184. https://doi.org/10.1038/NGEO2071
- Polyakov, I. V., Alkire, M. B., Bluhm, B. A., Brown, K. A., Carmack, E. C., Chierici, M., et al. (2020). Borealization of the Arctic Ocean in response to anomalous advection from Sub-Arctic seas. Frontiers in Marine Science, 7, 491. https://doi.org/10.3389/fmars.2020.00491

19448007, 2023, 4, Downloaded

- Polyakov, I. V., Pnyushkov, A. V., Alkire, M. B., Ashik, I. M., Baumann, T. M., Carmack, E. C., et al. (2017). Greater role for Atlantic inflows on sea-ice loss in the Eurasian Basin of the Arctic Ocean. *Science*, *356*(6335), 285–291. https://doi.org/10.1126/science.aai8204
- Polyakov, I. V., Rippeth, T. P., Fer, I., Alkire, M. B., Baumann, T. M., Carmack, E. C., et al. (2020). Weakening of cold halocline layer exposes sea ice to oceanic heat in the eastern Arctic Ocean. *Journal of Climate*, 33(18), 8107–8123. https://doi.org/10.1175/JCLI-D-19-0976.1
- Polyakov, I. V., Timokhov, L., Alexeev, V., Bacon, S., Dmitrenko, I., Fortier, L., et al. (2010). Arctic Ocean warming contributes to reduced polar ice cap. Journal of Physical Oceanography, 40(12), 2743–2756. https://doi.org/10.1175/2010JPO4339.1
- Rainville, L., & Woodgate, R. A. (2009). Observations of internal wave generation in the seasonally ice-free Arctic. Geophysical Research Letters, 36, L23604. https://doi.org/10.1029/2009GL041291
- Rantanen, M., Karpechko, A. Y., Lipponen, A., Nordling, K., Hyvärinen, O., Ruosteenoja, K., et al. (2022). The Arctic has warmed nearly four times faster than the globe since 1979. Communications Earth & Environment, 3(1), 168. https://doi.org/10.1038/s43247-022-00498-3
- Renssen, H., Seppä, H., Heiri, O., Roche, D. M., Goosse, H., & Fichefet, T. (2009). The spatial and temporal complexity of the Holocene thermal maximum. *Nature Geoscience*, 2(6), 411–414. https://doi.org/10.1038/ngeo513
- Riahi, K., van Vuuren, D. P., Kriegler, E., Edmonds, J., O'Neill, B. C., Fujimori, S., et al. (2017). The Shared Socioeconomic Pathways and their energy, land use, and greenhouse gas emissions implications: An overview. *Global Environmental Change*, 42, 153–168. https://doi. org/10.1016/j.gloenvcha.2016.05.009
- Screen, J. A., Deser, C., Smith, D. M., Zhang, X., Blackport, R., Kushner, P. J., et al. (2018). Consistency and discrepancy in the atmospheric response to Arctic sea-ice loss across climate models. *Nature Geoscience*, 11(3), 155–163. https://doi.org/10.1038/s41561-018-0059-y
- Serreze, M. C., & Barry, R. G. (2011). Processes and impacts of Arctic amplification: A research synthesis. *Global and Planetary Change*, 77(1), 85–96. https://doi.org/10.1016/j.gloplacha.2011.03.004
- Shu, Q., Wang, Q., Årthun, M., Wang, S., Song, Z., Zhang, M., & Qiao, F. (2022). Arctic Ocean Amplification in a warming climate in CMIP6 models. *Science Advances*, 8(30), eabn9755. https://doi.org/10.1126/sciady.abn9755
- Shu, Q., Wang, Q., Song, Z., & Qiao, F. (2021). The poleward enhanced Arctic Ocean cooling machine in a warming climate. Nature Communications, 12(1), 2966. https://doi.org/10.1038/s41467-021-23321-7
- Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. *Climate Dynamics*, 53(9–10), 5279–5291. https://doi.org/10.1007/s00382-019-04870-6
- Sirevaag, A., dela Rosa, S., Fer, I., Nicolaus, M., Tjernström, M., & Mcphee, M. (2011). Mixing, heat fluxes and heat content evolution of the Arctic Ocean mixed layer. Ocean Science, 7(3), 335–349. https://doi.org/10.5194/os-7-335-2011
- Skagseth, Ø., Eldevik, T., Årthun, M., Asbjørnsen, H., Lien, V. S., & Smedsrud, L. H. (2020). Reduced efficiency of the Barents Sea cooling machine. Nature Climate Change, 10(7), 661–666. https://doi.org/10.1038/s41558-020-0772-6
- Smedsrud, L. H., Ingvaldsen, R., Nilsen, J. E. Ø., & Skagseth, Ø. (2010). Heat in the Barents Sea: Transport, storage, and surface fluxes. Ocean Science, 6(1), 219–234. https://doi.org/10.5194/os-6-219-2010
- Smith, D., Screen, J., Deser, C., Cohen, J., Fyfe, J., García-Serrano, J., et al. (2019). The Polar Amplification Model Intercomparison Project (PAMIP) contribution to CMIP6: Investigating the causes and consequences of polar amplification. *Geoscientific Model Development*, 12(3), 1139–1164. https://doi.org/10.5194/gmd-12-1139-2019
- Steele, M., Ermold, W., & Zhang, J. (2011). Modeling the formation and fate of the near-surface temperature maximum in the Canadian Basin of the Arctic Ocean. Journal of Geophysical Research, 116, C11015. https://doi.org/10.1029/2010JC006803
- Steele, M., Morley, R., & Ermold, W. (2001). PHC: A global ocean hydrography with a high-quality Arctic Ocean. Journal of Climate, 14(9), 2079–2087. https://doi.org/10.1175/1520-0442(2001)014<2079:PAGOHW>2.0.CO;2
- Tsamados, M., Feltham, D. L., Schroeder, D., Flocco, D., Farrell, S. L., Kurtz, N., et al. (2014). Impact of variable atmospheric and oceanic form drag on simulations of Arctic sea ice. *Journal of Physical Oceanography*, 44(5), 1329–1353. https://doi.org/10.1175/JPO-D-13-0215.1
- Tsubouchi, T., Våge, K., Hansen, B., Larsen, K. M. H., Østerhus, S., Johnson, C., et al. (2021). Increased ocean heat transport into the Nordic seas and Arctic Ocean over the period 1993–2016. *Nature Climate Change*, 11(1), 21–26. https://doi.org/10.1038/s41558-020-00941-3
- Uotila, P., Goosse, H., Haines, K., Chevallier, M., Barthélemy, A., Bricaud, C., et al. (2019). An assessment of ten ocean reanalyses in the polar regions. *Climate Dynamics*, 52(3), 1613–1650. https://doi.org/10.1007/s00382-018-4242-z
- van der Linden, E. C., Le Bars, D., Bintanja, R., & Hazeleger, W. (2019). Oceanic heat transport into the Arctic under high and low CO₂ forcing. *Climate Dynamics*, 53(7), 4763–4780. https://doi.org/10.1007/s00382-019-04824-y
- Wang, Q., & Danilov, S. (2022). A synthesis of the upper Arctic Ocean circulation during 2000–2019: Understanding the roles of wind forcing and sea ice decline. *Frontiers in Marine Science*, 9, 863204. https://doi.org/10.3389/fmars.2022.863204
- Wang, Q., Wekerle, C., Wang, X., Danilov, S., Koldunov, N., Sein, D., et al. (2020). Intensification of the Atlantic water supply to the Arctic Ocean through Fram Strait induced by Arctic sea ice decline. *Geophysical Research Letters*, 47, e2019GL086682. https://doi.org/10.1029/ 2019GL086682
- Zelinka, M. D., Myers, T. A., McCoy, D. T., Po-Chedley, S., Caldwell, P. M., Ceppi, P., et al. (2020). Causes of higher climate sensitivity in CMIP6 models. *Geophysical Research Letters*, 47, e2019GL085782. https://doi.org/10.1029/2019GL085782
- Zuo, H., Balmaseda, M. A., Tietsche, S., Mogensen, K., & Mayer, M. (2019). The ECMWF operational ensemble reanalysis–analysis system for ocean and sea ice: A description of the system and assessment. *Ocean Science*, *15*(3), 779–808. https://doi.org/10.5194/os-15-779-2019